Author's response to reviews

Title: Using novel Canadian resources to improve medication reconciliation at discharge: study protocol for a randomized controlled trial

Authors:

  Robyn Tamblyn (robyn.tamblyn@mcgill.ca)
  Allen R Huang (allen.huang@muhc.mcgill.ca)
  Ari N Merguerditchian (ari.merguerditchian@mcgill.ca)
  Nancy E Winslade (nancy.winslade@mcgill.ca)
  Christian Rochefort (christian.rochefort@mcgill.ca)
  David Buckeridge (david.buckeridge@mcgill.ca)
  André Jacques (ajacques@cmq.org)
  Kiyuri Naicker (kiyuri.naicker@gmail.com)
  Kristen E Reidel (kristen.reidel@mail.mcgill.ca)
  Tewodros Eguale (tewodros.eguale@mail.mcgill.ca)
  Alan Forster (aforster@ohri.ca)

Version: 6 Date: 31 July 2012

Author's response to reviews:

July 31, 2012

Doug Altman
Editor-in-Chief, Trials

Dear Dr. Altman,

RE: Submission of revision for manuscript “USING NOVEL CANADIAN RESOURCES TO IMPROVE MEDICATION RECONCILIATION AT DISCHARGE: STUDY PROTOCOL FOR A RANDOMIZED CONTROLLED TRIAL” for publication by the journal Trials.

We were extremely pleased to hear of your decision to publish our above-mentioned manuscript in Trials. We have submitted a revision of the manuscript, so that the title format no follows what Trials requires, and we have also placed the Figure titles and legends and Tables sections at the end of the document, in order to follow the Trials format outlined in your instructions for authors:

>http://www.trialsjournal.com/authors/instructions/studyprotocol. Both the final version of manuscript and a version showing the changes using Track Changes have been uploaded to Trials.

Thank you for your time and consideration of this manuscript, we look forward to seeing it in print. If there are any further requirements, please do not hesitate to contact me.
Yours sincerely,
Robyn Tamblyn, PhD.
McGill Clinical and Health Informatics Research Group
1140 Pine Avenue West
Montreal, Quebec, H3A 1A3
Canada
Tel (514) 934 1934 x32999
Fax (514) 843 1551